Author's response to reviews

Title: Time trends in cardiovascular disease mortality in Russia and Germany from 1980 to 2007 - Are there migration effects?

Authors:

Volker Winkler (Volker.Winkler@ urz.uni-heidelberg.de)
Ari Paltiel (ari@cbs.gov.il)
Oliver Razum (oliver.razum@uni-bielefeld.de)
Heiko Becher (heiko.becher@urz.uni-heidelberg.de)
Andreas Deckert (deckert@imbi.uni-heidelberg.de)

Version: 2 Date: 17 March 2010

Author's response to reviews:

Dear Mrs. Norton,

during formatting the manuscript and adapting the figures to the requirements of BMC Public Health some unfortunate mistakes in naming the figures within the text occurred. Please find here today the newest version with the correct assignment of the figures in the text.

Kind regards

Andreas Deckert, corresponding author